Author's response to reviews

Title: Changes in body composition measured by mobile whole body MRI during a 4,486km transcontinental ultramarathon. Results from the TransEurope FootRace Project.

Authors:

Uwe HW Schütz (uwe.schuetz@rocketmail.com)
Jürgen Machann (Juergen.Machann@med.uni-tuebingen.de)
Kathrin König (ail.kathrinkoenig@googlemail.com)
Christian Würslin (Christian.Wuerslin@med.uni-tuebingen.de)
Heike Wiedelbach (heike.wiedelbach@uniklinik-ulm.de)
Hans-Jürgen Brambs (hans-juergen.brambs@uniklinik-ulm.de)
Christian Billich (christian.billich@uniklinik-ulm.de)

Version: 2 Date: 7 July 2012

Author's response to reviews:

Mistake in figure numbering. Correction was done.